



# Water Resources Research<sup>®</sup>

## **RESEARCH ARTICLE**

10.1029/2024WR039641

#### **Key Points:**

- The propagation from atmospheric water deficit to lake drought was enhanced at 28% of the selected lake basins in China
- Reservoirs are more susceptible to atmospheric water deficits than natural lakes
- Lake droughts were determined by lake-specific hydrological conditions at 37% of the selected lake basins

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

#### **Correspondence to:**

X. Liu, liuxm@igsnrr.ac.cn

#### Citation:

Zhang, D., Liu, X., Li, X., Wang, K., Fan, X., & Yuan, C. (2025). Propagation from atmospheric water deficit to lake drought in lake basins across China: Implications for lake drought management. *Water Resources Research*, *61*, e2024WR039641. https://doi.org/10.1029/ 2024WR039641

Received 8 DEC 2024 Accepted 18 MAY 2025

#### **Author Contributions:**

Conceptualization: Dan Zhang Investigation: Canyu Yuan Methodology: Dan Zhang Supervision: Xiaomang Liu Writing – original draft: Dan Zhang Writing – review & editing: Dan Zhang, Xiaomang Liu, Xianghu Li, Kaiwen Wang, Xingwang Fan

© 2025 The Author(s).

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited and is not used for commercial purposes.

## Propagation From Atmospheric Water Deficit to Lake Drought in Lake Basins Across China: Implications for Lake Drought Management

Dan Zhang<sup>1,2</sup>, Xiaomang Liu<sup>3</sup>, Xianghu Li<sup>1,2</sup>, Kaiwen Wang<sup>3</sup>, Xingwang Fan<sup>1,2</sup>, and Canyu Yuan<sup>1,2</sup>

<sup>1</sup>State Key Laboratory of Lake and Watershed Science for Water Security, Nanjing Institute of Geography and Limnology, Chinese Academy of Sciences, Nanjing, China, <sup>2</sup>Poyang Lake Wetland Research Station, Nanjing Institute of Geography and Limnology, Chinese Academy of Sciences, Jiujiang, China, <sup>3</sup>Institute of Geographic Sciences and Natural Resources Research, Chinese Academy of Sciences, Beijing, China

**Abstract** Global lakes are expected to witness a growing incidence of droughts under climate warming, but the propagation from atmospheric water deficit (AWD) to lake drought, mediated by basin water deficit (BWD) as a key intermediary, remains underexplored at the lake basin scale. This study investigated the propagation from AWD to lake drought in 1,617 lake basins across China using a copula-based approach. 19% of AWD events cascaded to lake droughts via BWD, while 57% of lake droughts were traceable to antecedent AWDs. These results underscore the significance of basin modulation in determining the responses of lake droughts to climate variability. Notably, the propagation ability for reservoir basins was higher than that of natural lake basins, indicating that reservoirs are more susceptible to AWDs than natural lakes. A classification framework for China's lake basins was further proposed for lake drought management. 28% of lake basins exhibited lake drought intensification through the AWD-to-BWD path, necessitating climate-adaptive measures. 6% of lake basins faced lake drought intensification driven by BWD with no AWD occurrence, requiring integrated lake basin management. 18% of basins with lake drought intensification were controlled by lake-specific hydrological conditions, demanding localized strategies to maintain water stability. The remaining lake basins exhibited a reduced probability of lake drought occurrence. This study provides actionable guidance to tailor lake drought management, prioritizing climate-vulnerable basins or lake-specific interventions.

**Plain Language Summary** Although lakes cover only 3% of the global land area, they are essential to the global hydrological and biogeochemical cycles. Recently, an increasing number of lakes have suffered droughts under climate warming, yet the propagation from atmospheric water deficit to lake drought remains poorly understood at large scales. Lake drought is initially triggered by an atmospheric water deficit and further modulated by basin hydrological processes. Few studies have focused on hydrological propagation in specific lake basins using hydrological-hydrodynamic models, which are highly data-intensive and time-consuming. In this study, we investigated the propagation chain from atmospheric water deficit to basin water deficit to lake drought in 1,617 lake basins across China using a copula-based approach. Then, a lake basin classification framework for lake drought management was proposed according to the changes in this propagation chain. The results of this study provide decision support for prioritizing lake basins requiring climate-adaptive policies and lake-specific interventions.

## 1. Introduction

Although lakes cover only 3% of the global land area, they harbor a substantial portion of terrestrial water resources, which are crucial for socio-economic development and ecosystem sustainability (Weyhenmeyer et al., 2024; Yao et al., 2023). In a warming climate, these functions are anticipated to be at risk from the increasingly frequent lake drought, which is identified by combining a specific threshold with the lake water deficit status, as indicated by a drop in water level, surface area, or storage below normal conditions (Woolway et al., 2020). Generally, lake drought is initially triggered by an atmospheric water (precipitation) deficit, which subsequently propagates to a lake water deficit through basin hydrological processes, and can be further intensified by human water withdrawal (Cooley et al., 2021; Perales et al., 2020; Zhang, Li, et al., 2023). The propagation from atmospheric water deficit (AWD) to lake drought reflects water shortage propagation processes from the atmosphere to the land and from the basin to the lake, which directly affect the water availability, nutrient concentration, and ecological health of lakes (Saber et al., 2020; Tong et al., 2023). Yet, this important process of atmosphere–basin–lake water shortage propagation has received little attention from limnologists, hydrologists, meteorologists, and remote sensing scientists.

Few limnologists have investigated the propagation from AWD to lake drought in specific large lake basins and lake basins that significantly impact local socioeconomics using hydrological models in combination with lake hydrodynamic models (Kummu et al., 2014; Lai et al., 2013). Such a process-based method can depict water transfer processes in lake basins at different time scales, but exhibits two disadvantages. On one hand, this method is highly data-intensive, with its accuracy heavily dependent on precise lake bathymetry (Yao et al., 2018). Despite the rapid development of remote sensing technology, retrieving underwater bathymetry remains a challenge (Liu & Song, 2022; Luo et al., 2022). On the other hand, the lake model needs to be customized and the computation is time-consuming. These limitations make the method impractical for large-scale studies. Instead, statistical approaches can circumvent these disadvantages, including correlation analysis, machine learning, and copula-based approach, of which the copula-based approach has been used to model nonlinear dependencies between continuous hydro-meteorological variables (Gaupp et al., 2020; Jiang et al., 2023; Qing et al., 2023). For example, it is assumed that there is a propagation from meteorological drought to hydrological drought when the two types of droughts co-occurred (Zhang et al., 2017). Moreover, the copula-based approach overcomes the shortcoming of assessing the relationship between hydro-meteorological extremes with few samples, making it possible to investigate hydrological propagation relationships in lake basins at any other meteorological drought and the samples, making it possible to investigate hydrological propagation relationships in lake basins with limited observational data.

Hydrology and remote sensing researchers are concerned with the dynamics of hydrological variables in lakes, such as lake water area, storage, and depth, but the research on hydrological extremes of lakes and their connections with atmospheric conditions through basin modulation remains underexplored (Cooley et al., 2021; Pekel et al., 2016; Tong et al., 2023). For example, Yao et al. (2023) constructed time-varying water storage in 1,972 global largest lakes from 1992 to 2020, and the relationships between lake water storage and precipitation were only subsidiary outcomes. Xu et al. (2024) found that the projected lake expansion in the Tibetan Plateau is primarily fueled by amplified water inputs from increased precipitation and glacier meltwater, profoundly reshaping the hydrological connectivity of lake basins. These studies focused on the inversion techniques and approaches of lake variables. As a crucial hydrological relationship in lake basins, the propagation from AWD to lake drought through basin water deficit (BWD) reflects the water shortage transfer processes in atmosphere–basin–lake systems, emphasizing the role of atmospheric forcing and basin-scale processes in driving lake droughts. This propagation is initiated by atmospheric forcing and modulated by basin- and lake-scale processes. Understanding such hydrological propagation mechanisms is a prerequisite for integrated lake basin management to achieve effective drought governance (Awange et al., 2008; Soeprobowati, 2015).

In this light, this study takes 1,617 lake basins of China as an example and combines satellite products and a copula-based approach to: (a) analyze the spatial patterns of propagation from AWD to lake drought, (b) investigate the temporal changes in the propagation from AWD to lake drought during the past 40 years, and (c) propose a lake basin classification framework for effective lake drought management. The results are of great significance for understanding the hydrological transfer mechanism in atmosphere–basin–lake systems, which are expected to assist the lake drought management in lake basins under climate change.

## 2. Methods and Materials

#### 2.1. Methods

### 2.1.1. Definitions of AWD, BWD, and Lake Drought

Many definitions of drought and water deficit have been proposed in previous studies, among which the standard precipitation index (SPI) has been widely used and extended to other hydrological variables, such as river runoff, soil moisture, groundwater levels, and lake water area (AghaKouchak et al., 2021; Guttman, 1999; McKee et al., 1993). The SPI usually employs the Gamma distribution to calculate the cumulative probability distribution of precipitation F(x):





$$F(x) = \begin{cases} \int_0^x \frac{t^{\alpha-1}e^{-t/\beta}}{\beta^{\alpha}\Gamma(\alpha)} dt & (x>0) \\ N_{zero}/N_{total} & (x=0) \end{cases}$$
(1)

where x represents precipitation,  $\alpha$  is the shape parameter,  $\beta$  is the scale parameter,  $\Gamma$  is the Gamma function,  $N_{zero}$  is the number of months with zero precipitation, and  $N_{total}$  is the total number of months. F(x) is then transformed into the standard normal distribution quantile to derive the SPI value.

Similarly, the Standard Runoff Index (SRI) and Standard lake Water area Index (SWI) were employed to quantify basin and lake water shortage (Shukla & Wood, 2008; Zhang et al., 2017). This study utilized 12-month-scale SPI, SRI, and SWI to assess AWD, BWD, and lake drought, respectively. Lake drought was defined as SWI  $\leq -1$ , while AWD and BWD were identified when SPI and SRI values fall below zero, indicating water deficit conditions (McKee et al., 1993). The SWI threshold of -1 follows standardized index conventions such as the SPI, corresponding to one standard deviation below the mean (15.9% cumulative probability). This ensures crossvariable drought assessment consistency (e.g., precipitation, runoff). This study defines AWD and BWD with thresholds of 0 (SPI  $\leq$  0, SRI  $\leq$  0), contrasting with conventional drought thresholds (SPI  $\leq -1$ , SRI  $\leq -1$ ), to fully characterize basin-mediated water shortage propagation. Even under non-drought atmospheric and basin conditions (SPI > -1, SRI > -1), persistent AWD and BWD can trigger lake droughts through cumulative hydrological feedbacks—a process traditional drought-focused methods often overlook. By extending these definitions, the approach enhances detection of AWD–BWD–lake drought dynamics in lake basins.

Moreover, the Gamma, Generalized Extreme Value (GEV), and Pearson Type III distributions were applied to fit the precipitation, runoff, and lake water area series using L-moment estimation, aiming to accurately characterize the probability distributions of these variables (Wang et al., 2021; Xu et al., 2015). If a distribution passes the Kolmogorov-Smirnov test at the 95% confidence level, the candidate distributions are further compared using the Bayesian Information Criterion (BIC) (Kole et al., 2007; Neath & Cavanaugh, 2012).

#### 2.1.2. Calculation of Propagation Probability From AWD to Lake Drought

Within the Copula framework, we assumed a propagation from AWD to lake drought when the AWD, BWD, and lake drought coincide in a given lake basin (Figure 1a). The concurrent probability ( $P_{vine}$ ) of AWD, BWD, and lake drought was estimated using a copula method based on Sklar's theorem (Sklar, 1959):

$$P_{vine}(\text{SPI} \le 0, \text{SRI} \le 0, \text{SWI} \le -1) = C[F_{\text{SPI}}(0), F_{\text{SRI}}(0), F_{\text{SWI}}(-1)]$$
(2)

where  $P_{vine}$  is concurrent probability of SPI  $\leq 0$ , SRI  $\leq 0$ , and SWI  $\leq -1$ ,  $F_{SPI}$ ,  $F_{SRI}$ , and  $F_{SWI}$  are the marginal probability distributions of SPI, SRI, and SWI, and  $C \rightarrow [0, 1]$  is a copula function. Regular vine tree structures were employed to model the complex dependencies of the SPI, SRI, and SWI (Bedford & Cooke, 2002). For each copula pair, copula families with a single parameter of Kendall's tau were employed, including Gaussian, Clayton, Gumbel, Frank, and Joe copulas. The maximum likelihood estimation was used to calculate the parameter for each bivariate copula and the BIC was used to select the optimal copula model for each copula pair. Notably,  $P(AWD \cap LDT)$  in Figure 1a doesn't represent the propagation from AWD to lake drought. This is because the basin serves as the critical intermediary linking atmospheric and lake processes. When AWD occurs, it first alters basin hydrological dynamics—such as reducing runoff and soil moisture. These changes subsequently impact lake hydrology through surface and subsurface flow pathways, forming a cascading atmosphere–basin–lake water shortage response. If AWD is not accompanied by BWD, lake drought in such cases is likely dominated by internal hydrological processes, such as excessive evaporation or anthropogenic outflow regulation.

According to the graphical relationship in Figure 1a, the conditional probability of  $P_{vine}$  given AWD significantly differs from that given lake drought, indicating two critical insights: (a) not all AWDs propagate to lake droughts due to basin-scale buffering effects, and (b) not all lake droughts originate from AWDs (e.g., anthropogenic withdrawals). To disentangle these dual dynamics, two probabilistic metrics were formalized:





Figure 1. (a) Venn diagram of propagation from AWD to lake drought. (b) Conceptual classification framework of lake basins based on propagation ability from AWD to lake drought.

$$P_{atmos \to lake}(BWD \cap LDT | AWD) = \frac{P_{vine}}{P_{AWD}}$$
(3)

$$P_{atmos|lake}(AWD \cap BWD|LDT) = \frac{P_{vine}}{P_{LDT}}$$
(4)

where  $P_{AWD}$  and  $P_{LDT}$  are the probabilities of AWD (SPI  $\leq 0$ ) and lake drought (SWI  $\leq -1$ ), respectively, estimated by the marginal probability distributions using the GEV.  $P_{atmos \rightarrow lake}$  is the conditional probability of the simultaneous occurrence of BWD and lake drought given the occurrence of AWD, representing the likelihood of AWD propagating to lake drought via BWD.  $P_{atmos|lake}$  is the conditional probability of the simultaneous occurrence of AWD and BWD given the occurrence of lake drought, representing the proportion of lake drought events linked to AWD. These two propagation probabilities were used to evaluate the propagation ability of water shortage transfer within the atmosphere–basin–lake system. Higher values indicate stronger atmosphere–land coupling in the lake basin.



Similarly, the propagation probabilities from AWD to BWD and from BWD to lake drought were estimated to explore the water shortage transfer processes from the atmosphere to the basin and hence to the lake, calculated as follows:

$$P_{atmos \to basin}(BWD|AWD) = \frac{P(AWD \cap BWD)}{P_{AWD}}$$
(5)

$$P_{atmos|basin}(AWD|BWD) = \frac{P(AWD \cap BWD)}{P_{BWD}}$$
(6)

$$P_{basin \to lake}(\text{LDT}|\text{BWD}) = \frac{P(\text{BWD} \cap \text{LDT})}{P_{\text{BWD}}}$$
(7)

$$P_{basin|lake}(BWD|LDT) = \frac{P(BWD \cap LDT)}{P_{LDT}}$$
(8)

where  $P_{BWD}$  is the probability of BWD (SRI  $\leq 0$ ),  $P(AWD \cap BWD)$  and  $P(BWD \cap LDT)$  are the concurrent probabilities of AWD-BWD and BWD-lake drought, as illustrated in Figure 1a.

The changes in the propagation probabilities were calculated as the difference between the changed period 2000–2018 and the whole period 1985–2018 (see details in the Materials). For SPI, SRI, and SWI calculations in each period, distribution parameters fitted over the whole period were used to characterize water deficit and drought changes. This study selected the period 1985–2018 for probabilistic model fitting to enhance analytical reliability. Given the lower frequency of remote sensing observations during 1985–1999, the reconstructed monthly lake water area data from this earlier period may contain greater uncertainties in capturing extreme lake drought events. By maintaining original data standards (1985–2018), the increased effective sample size improves the statistical representativeness of hydrological time series. The whole period better captures hydrological variability characteristics, thereby enhancing the stability of probability distribution fitting for lake water area and reducing parameter estimation biases caused by small sample sizes.

#### 2.1.3. Lake Basin Classification for Lake Drought Management

This study proposes a novel lake basin classification framework for lake drought management based on water shortage propagation within the atmosphere–basin–lake system. Lake basins are classified into three groups with six subtypes through a three-tiered decision process (Figure 1b). This process evaluates directional consistency of the changes in propagation probabilities (e.g.,  $P_{atmos \rightarrow lake}$  and  $P_{atmos]lake}$ ) in each lake basin.

- Group 1 (Atmosphere–Basin–Lake Linkage): a lake basin is assigned to Type I (increased propagation ability
  from AWD to lake drought via BWD) if both P<sub>atmos→lake</sub> and P<sub>atmos|lake</sub> exceed certain thresholds (set 0.1 and
  0.3 in this study, respectively), and exhibit positive changes. Conversely, if both probabilities show negative
  changes, it is classified as Type II (decreased propagation ability from AWD to lake drought via BWD). This
  group highlights atmospheric forcing mediated through basin processes.
- Group 2 (Basin–Lake Linkage): Unclassified lake basins are re-evaluated using the P<sub>basin→lake</sub> and P<sub>basin|lake</sub>. Type III (increased propagation ability from BWD to lake drought) or Type IV (decreased propagation ability from BWD to lake drought) is assigned based on positive or negative changes. This group emphasizes basin-regulated propagation independent of AWD.
- Group 3 (Lake-Dominant Dynamics): Remaining lake basins are classified as Type V (increased lake drought) or Type VI (decreased lake drought) based on changes in P<sub>LDT</sub>. Lake drought dynamics here are governed by lake-specific factors (e.g., evaporation, outflow regulation) without BWD.

This classification framework delineates AWD–BWD–lake drought propagation pathways (Figure 1b), providing a decision-support tool for prioritizing lake basins requiring climate-adaptive policies (e.g., Type I) versus lake-specific interventions (e.g., Type V).

#### 2.2. Materials

The lake data of China were obtained from the HydroLAKES dataset (Messager et al., 2016). 1,617 lakes were chosen to investigate the propagation from AWD to lake droughts subject to the following selection criteria. First,





Figure 2. Sketch map of the selected lakes. The points are the lake outlets in their basins.

the basin area is larger than 100 km<sup>2</sup> and the lake area is larger than 1 km<sup>2</sup> to ensure a hydrological propagation from the basin to the lake. Second, the marginal distribution of each variable passes the Kolmogorov-Smirnov test at the 95% confidence level. Third, the  $P_{AWD}$ ,  $P_{BWD}$  and  $P_{LDT}$  are greater than 0.01 (return period of one-hundredyear) for the probability below this threshold might lead to unreliable results. The area of selected lakes ranges from 1 to 4,267 km<sup>2</sup>, with a total coverage of 55,255 km<sup>2</sup>, accounting for 56.8% of China's total lake area. The basin area ranges from 100 to 657,165 km<sup>2</sup>. This study did not adopt pre-defined sub-basins from HydroBASINS due to their inconsistency with actual lake drainage boundaries. Instead, an automated watershed delineation approach was employed to independently extract the basin area for each lake by treating the lake pour point as the basin outlet (Xie et al., 2022). This refined delineation accurately characterizes the hydro-connectivity topology of the lake and its basin, providing reliable geographic units to investigate the hydrological transport processes in the atmosphere–basin–lake system. The ratio of basin area to lake area ranges from 1.3 to 341,802. The selected 1,617 lakes were classified as natural lakes and reservoirs according to a comprehensive reservoir dataset (CRD) of China released by Song et al. (2022b). For a certain lake of HydroLAKES, the boundaries of HydroLAKES and CRD overlapped first. When the overlapping area exceeded 70% of HydroLAKES, the lake type was assigned as reservoirs. The other lakes were assigned to natural lakes. Finally, 882 natural lakes and 735 reservoirs were classified, accounting for 83% and 17% of the total area of selected lakes, respectively (Figure 2). All the lakes were categorized into six zones according to the geographical division of China, including Yunnan-Guizhou Plateau (YGP), Tibetan Plateau (TP), Uygur Autonomous Region (UAR), Inner Mongolia Plateau (IMP), Northeast Plains and Mountains (NEPM), and Eastern Plains (EP).

The monthly lake water area from 1985 to 2018 was reconstructed by Zhao et al. (2022), using a robust image enhancement algorithm based on the dynamic Landsat-based global surface water dataset and the static HydroLAKES boundaries. The data spanned from 1985 to 2018, that is, the whole observation period. The period from 2000 to 2018 was considered as the changed period due to intensified human activities and high-frequency satellite observations. The precipitation and runoff (sum of surface and subsurface runoffs) of each lake basin were obtained from the Inter-Sectoral Impact Model Intercomparison Project phase 3a (ISIMIP3a) (Frieler et al., 2024). Runoff from five terrestrial hydrology models was employed, including four global hydrological models (GHMs; CWatM, H08, HydroPy, and WaterGAP2-2e) and one global land surface model (MIROC-INTEG-LAND). The spatial resolution of ISIMIP data is  $0.5^{\circ} \times 0.5^{\circ}$ . The population and Gross Domestic Product (GDP) datasets of China in 2015 with a resolution of 1 km × 1 km were employed to estimate the impacts of propagation from AWD to lake drought on socioeconomic systems (Wang & Wang, 2022a; Wang





Figure 3. Spatial distributions of propagation probabilities (a)  $P_{atmos \rightarrow lake}$  and (b)  $P_{atmos|lake}$  from AWD to lake drought in China. Changes in (c)  $P_{atmos \rightarrow lake}$  and (d)  $P_{atmos|lake}$ , calculated as the difference between each variable during 2000–2018 and 1985–2018. The points are the lake outlets of lake basins. The inner and outer rings of the donut represent the proportions of the lake number and area to the total values, respectively.

& Wang, 2022b). The land use dataset of China in 2015 with a resolution of 1 km  $\times$  1 km was used to analyze the impacts of different land use types on the propagation. A lake basin was classified as cropland-, forest-, or grassland-dominated if any single land cover type exceeded 50% of its total area. In addition, lake basins with >10% residential land cover was categorized as residential-influenced, reflecting urbanization impacts on water shortage propagation.

## 3. Results

## 3.1. Distributions of Propagation Probabilities From AWD to Lake Drought

The propagation probabilities from AWD to lake drought calculated from the five terrestrial hydrology models were first compared (Figure S1 in Supporting Information S1). Equations 3 and 4 demonstrate that differences in propagation probabilities between models primarily stem from  $P_{vine}$  variations, that is, the concurrent probability of AWD, BWD, and lake drought. Figure S1 in Supporting Information S1 shows that the  $P_{vine}$  from different models correlates well, with Pearson correlation coefficients exceeding 0.78. Therefore, it is reasonable to use the average values from the five models to reduce uncertainties in propagation probability calculations. Figures 3a and 3b show the spatial distributions of propagation probabilities from AWD to lake drought in China's lake basins. Generally, high propagation probabilities mainly occurred in the Tibetan Plateau and the low values occurred in Northwestern China. The  $P_{atmos \rightarrow lake}$  averaged from the selected lake basins was  $0.19 \pm 0.06$  (one standard deviation), meaning that 19% of AWDs propagated to lake droughts at the national scale. The  $P_{atmos|lake}$  averaged from the selected lake basins were related to AWDs. The  $P_{atmos|lake}$  values were larger than  $P_{atmos \rightarrow lake}$  because the  $P_{LDT}$  values were smaller than the  $P_{AWD}$ . Among the six lake zones, the high propagation probabilities occurred in the TP and YGP zones during 1985–2018, as shown by blue bars in Figures 4a and 4b. The high propagation probabilities in these two zones are primarily driven by intense land-atmosphere coupling characteristic of high-altitude regions, where strong





**Figure 4.** Propagation probabilities (a)  $P_{atmos \rightarrow lake}$ , and (b)  $P_{atmos|lake}$  from AWD to lake drought for the six lake zones during 1985–2018 and 2000–2018. The black bar is one standard deviation. Comparison of propagation probabilities between natural lakes and reservoirs (c) from 1985 to 2018 and (d) from 2000 to 2018. On each box, the central mark is the median, the square is the average, the edges of the box are the 25th and 75th percentiles, and the whiskers are the most extreme values.

interactions between surface hydrology and atmospheric dynamics amplify water shortage transmission to lakes (Long et al., 2014; Yang et al., 2025; Zhu et al., 2023).

It is worth noting that reservoir basins exhibited higher propagation probabilities compared to natural lake basins during 1985–2018, indicating their greater susceptibility to AWDs (Figure 4c). This divergence primarily stems from contrasting hydrological regulation mechanisms: reservoirs experience accelerated water level fluctuations due to anthropogenic controls (e.g., irrigation releases, hydropeaking), which degrade their natural buffering capacity against AWDs, while natural lakes maintain gradual water level dynamics through delayed groundwater recharge and snowmelt-dominated hydrological cycles, with wetland ecosystems further attenuating water shortage propagation signals (Hughes et al., 2012; Xi et al., 2021). In addition, land use patterns in lake basins also influence the propagation probabilities from AWD to lake drought. As shown in Figure 5, the forest-dominated lake basins exhibit the highest drought propagation probability, followed by cropland- and grassland-dominated lake basins, while residential-influenced lake basins show the lowest probability. This difference arises from distinct hydrological regulation mechanisms across ecosystems: the strong evapotranspiration from forest canopies and soil moisture depletion caused by deep root systems intensifies the cascading transmission of atmospheric water shortage to soil and lake systems at the annual scale (Peterson et al., 2021; Yang et al., 2017). In contrast, cropland irrigation and grasslands' shallow water retention partially mitigates drought signal accumulation, creating moderate buffering effects. Conversely, impervious surfaces in residential areas promote rapid runoff generation, and artificial drainage systems disrupt natural hydrological connectivity, thereby weakening AWD-BWD-lake drought cascading effects (Rachunok & Fletcher, 2023; Salvadore et al., 2015).

The propagation processes were further disentangled by comparing propagation probabilities from the atmosphere to the basin and from the basin to the lake (Figures 6a and 6b). About 87% of AWDs converted to BWDs whereas only 21% of BWDs converted to lake droughts, indicating that AWDs strongly influenced BWDs,





**Figure 5.** Propagation probabilities (a)  $P_{atmos \rightarrow lake}$ , and (b)  $P_{atmos|lake}$  from AWD to lake drought for different land use types during 1985–2018. On each box, the central mark is the median, the square is the average, the edges of the box are the 25th and 75th percentiles, and the whiskers are the most extreme values.

whereas BWDs had a weaker influence on lake droughts at the annual scale. To reduce the impact of different water deficit thresholds of the atmosphere, the basin, and the lake, we set these three thresholds to the same value of -1 (Figures 6c and 6d). The propagation probabilities from AWD to BWD ( $P_{atmos} \rightarrow basin$ ,  $P_{atmos|basin}$ ) were (0.70  $\pm$  0.14, 0.72  $\pm$  0.14), whereas the propagation probabilities from BWD to lake drought ( $P_{basin \rightarrow lake}$ ,  $P_{basin|lake}$ ) were (0.30  $\pm$  0.14, 0.27  $\pm$  0.13). The results confirm that the propagation from AWD to BWD is robust, whereas the propagation from BWD to lake drought is significantly weaker, reflecting the complexity of lake drought dynamics.

## 3.2. Changes in Propagation Probabilities From AWD to Lake Drought

Figures 3c and 3d show the changes in  $P_{atmos \rightarrow lake}$  and  $P_{atmos|lake}$  in lake basins across China during 2000–2018 compared with 1985–2018. The changes in these two propagation probabilities had distinct regional characteristics. The positive values occurred in Northeast China and the south of 30–31°N, whereas the negative values occurred in the Tibetan Plateau and the plains of the Yellow, Huai, and Hai rivers. Although the spatial patterns of changes in  $P_{atmos} \rightarrow lake$  and  $P_{atmos|lake}$  are similar, the correlation coefficient between them is only 0.28, which is caused by the low relationship between  $P_{AWD}$  and  $P_{LDT}$  (Figure S2 in Supporting Information S1). The changes in  $P_{LDT}$  explained 81% of the changes in  $P_{atmos} \rightarrow lake$  and the changes in  $P_{AWD}$  explained 22% of the changes in  $P_{atmos|lake}$  (Figure S3 in Supporting Information S1). For the six zones, the changes in the propagation probabilities increased in the IMP, NEPM, EP, and YGP zones, indicating widespread enhanced propagation ability from AWD to lake drought in lake basins over the eastern regions of China (Figures 4a and 4b). In addition, the changes in propagation probabilities for the reservoir basins increased, indicating that the propagation from AWD to lake drought was enhanced for the reservoir basins (Figure 4d).



**Figure 6.** Propagation probabilities (a) from AWD to BWD and (b) from BWD to lake drought. (c)–(d) are similar to (a)–(b), but the water deficit thresholds of the atmosphere, the basin, and the lake were set to the same value of -1, that is, SPI  $\leq -1$ , SRI  $\leq -1$ , and SWI  $\leq -1$ .

#### 3.3. Lake Basin Classification for Lake Drought Management

The selected lake basins were classified into three groups with six subtypes based on our proposed framework (Figure 7a). 28% of the selected lake basins belonged to Type I, mainly located in northeast China and the middlelower reaches of the Yangtze River. The propagation ability from AWD to lake drought in these lake basins increased, resulting in a risk of lake shrinkage. Specifically, the lake basins in northeast China need to be alert to the risk of water scarcity, and the lake basins in the middle-lower reaches of the Yangtze River need to be alert to the crisis of ecosystem degradation (Cai et al., 2016; Li et al., 2024). These lake basins require an urgent response to address the adverse effects of climate change. Notably, Type I accounts for 33% (by number) of reservoir basins and 25% (by number) of natural lake basins (Figure 8), confirming that reservoirs are more susceptible to AWDs than natural lakes. 25% of the selected lake basins belong to Type II, which were mainly located in the Tibetan Plateau and the plains of the Yellow, Huai, and Hai rivers. The propagation ability from AWD to lake drought in these lake basins decreased. Taking the Tibetan Plateau as an example, lake basins in this region have mainly faced the problem of lake expansion in recent years (Xu et al., 2024). The propagation chain of Type I and Type II lake basins was AWD-BWD-lake drought. The increased (Type III) and decreased (Type IV) propagation ability from BWD to lake drought accounted for small proportions of the selected lake basins. Lake droughts of these two types were mainly related to BWDs, indicating that integrated lake basin management would be effective. The propagation chain of Type III and Type IV was non-AWD-BWD-lake drought. 18% and 19% of the selected lake basins belonged to Type V and Type VI, respectively. These lake basins were widely distributed throughout China. Drought changes in these lakes were predominantly governed by lake-specific hydrological conditions rather than basin-scale processes. These lake basins need to pay more attention to the impact of human activities or natural changes on the lakes themselves, as the basins' influence on lakes is limited.





Figure 7. (a) Lake basin classification across China. The inner and outer rings of the donut represent the proportions of the lake number and area of each type to the total values, respectively. (b) Types for the top 10 largest natural lakes and reservoirs according to the water area of the selected lake basins. The water area of the natural lakes and reservoirs from top to bottom decreases.

## 4. Discussion

## 4.1. Implications for Lake Drought Management

Although integrated lake basin management toward effective water governance has been advocated in many countries and organizations, this issue has not received sufficient attention in global water policies (Avra-moski, 2004; Cheng & You, 2019; Soeprobowati, 2015). China has proposed integrated lake basin management since 2017, named "a holistic approach to conserving mountains, rivers, forests, farmlands, lakes, and grasslands"



(a) Natural lakes number proportion

Figure 8. Classification heatmaps of the proportions of the lake number of each type to the total values for (a) natural lake basins and (b) reservoir basins. (c) And (d) are similar to (a) and (b), but for the proportions of lake area of each type to the total values.

(Cheng & You, 2019; Kong et al., 2019). According to our proposed lake basin classification framework, this study divided lake basins into three groups with six subtypes based on natural physical process of water shortage propagation in the atmosphere-basin-lake system, which provides new insights into effectively managing lake droughts in China.

Lake droughts in Type I lake basins were aggravated by AWD variations, reminding the decision-makers to be aware of lake shrinkage risks caused by atmospheric and runoff deficits. Figure 9 shows that Type I lake basins influenced 21% of the national land area, 16% of the population, and 13% of the GDP. In other words, the strong human interventions reflected by the population and GDP might also exacerbate the crisis of lake drought, such as irrigation, domestic water use, and reservoir operation (Apurv & Cai, 2021; Zhang et al., 2018). For example, Poyang Lake (Type I; Figure 7b), the largest freshwater lake in China, experienced record-breaking droughts in the summer of 2022 due to abnormal atmospheric precipitation (Zhang, Xue, & Xia, 2023). The operation of the Three Gorges Dam (TGD) exacerbated this drought, as it down-cut river channel and decreased runoff of the mainstream Yangtze River (Lai et al., 2014). Although the TGD has a function of drought alleviation, its regulation effect on this drought was limited due to the water released before the flood season. In addition, there are very high-density reservoirs in the Poyang Lake basin, but their main functions are flood control, power generation, and irrigation. It is necessary to strengthen the medium- and long-term hydrological forecasting capabilities for rivers and lakes and improve the regulating function of reservoirs during drought. From broader geophysical and climatic perspectives, reduced lake area and water storage may weaken evaporative cooling effects, thereby exacerbating regional heatwaves and altering local hydrological cycles (Fan et al., 2024; Wang et al., 2024). Such feedbacks could amplify atmosphere-land interactions, particularly in densely populated basins where human activities intersect with natural processes (Woolway et al., 2018). However, current Earth system models often oversimplify these coupled dynamics, limiting their capacity to predict cascading hydroclimatic impacts.





Figure 9. Proportions of area, population, and GDP affected by different lake basin types relative to national totals.

Compared to Type I lake basins, lake drought caused by AWD has been mitigated in Type II lake basins. Most of these lake basins are in the TP zone, where drought mitigation is primarily attributed to the increased precipitation resulting from climate warming (Xu et al., 2024). Lake drought management in these basins can be temporarily de-prioritized. However, the risk of lake shrinking in the other lake basins is still high. For instance,  $P_{LDT}$  in Bosten Lake (Type II; Figure 7b), the largest freshwater lake in Northwest China, increased from 0.14 in 1985-2018 to 0.22 in 2000-2018, reflecting decoupled atmospheric forcing and heightened anthropogenic impacts. The management strategy for these lake basins aligns with that of Type V lake basins, as lake droughts in Type V are predominantly governed by lake-specific hydrological conditions, including precipitation, evaporation, outflow regulation, and consumptive water use, rather than basin-scale processes. Hulun Lake, Hungtze Lake, and Gaoyou Lake belonged to this type (Figure 7b). Drought changes in these three lakes were mainly affected by human activities, and it is urgent to prevent water quality deterioration and ecosystem degradation (Cui & Li, 2016; Huang et al., 2023). In addition, attention should be paid to the drought changes in the Miyun Reservoir, which is the water destination

region of the central South–to–North Water Diversion project (Long et al., 2020). Although the water area of this reservoir has been recovered since 2015 due to the Water Diversion project, climate-driven risks require proactive mitigation (Zheng et al., 2016) (Figure S4 in Supporting Information S1).

#### 4.2. Limitations

The lake water area data used in this study have limitations that require future improvements. These data were reconstructed from remote sensing imagery (Zhao, 2021). Pre-2000 Landsat images suffered from low observation frequency and contamination by clouds, cloud shadows, and terrain shadows, resulting in substantial data gaps (Pekel et al., 2016; Vicente-Serrano et al., 2008). These gaps may underestimate lake hydrological fluctuations, such as rapid expansion following intense precipitation events. For example, Zhao and Gao (2018) demonstrated that correcting contaminated images for 6,817 global reservoirs (1984–2015) increased estimated reservoir surface area from  $1.73 \times 10^5$  km<sup>2</sup> to  $3.94 \times 10^5$  km<sup>2</sup>. In addition, the lake water area extraction relies on the buffered HydroLAKES database's static water body boundaries, but climate warming-driven lake expansion and basin reorganization have caused significant water area changes in high-altitude regions like the Qinghai-Tibet Plateau (Liu et al., 2021; Vacherat et al., 2018; Xu et al., 2024), potentially introducing inaccuracies in maximum lake water area estimates for these reorganized basins. Up to now, most lake modules in ecological, hydrological, water quality, and Earth system models parameterize lakes as static entities with fixed sizes (Frieler et al., 2024; Müller Schmied et al., 2023). This oversimplification severely constrains predictive capabilities for future evolution trends of large-scale hydrological processes in lake basins.

Second, the study analyzed the propagation from AWD to lake drought using SPI and SWI over the same 12month period, potentially underestimating the complexity of water shortage propagation in atmosphere–basin– lake systems (Sattar et al., 2019; Yang et al., 2024). We calculated correlation coefficients between SPI and SWI with lags of 1–12 months. Results show a maximum average correlation of 0.29 (Figure S5 in Supporting Information S1), explaining only 8% of SWI variability. This indicates that lagged effects were weak compared to dominant drivers like groundwater recharge and human activities. At annual or multi-year scales, SWI-SPI lagged relationships are masked by interacting hydrological processes (e.g., vegetation transpiration delays, irrigation withdrawals), which may generate spurious correlations (Peterson et al., 2021). This study used a probability statistical method focused on annual-scale analysis to avoid overinterpreting noise-driven signals, providing a robust framework for lake drought assessment.

Third, the lake basin classification did not consider the lake outflow due to the lack of observation records. The lake outflow is influenced by the hydrological conditions of the river where the lake flows, which cannot be ignored in some lake basins. Taking Poyang Lake, flowing into the Yangtze River, as an example, the decrease in water level of the Yangtze River leads to an increase in lake outflow and hence aggravates lake drought (Zhang et al., 2017). Exploring lake outflow dynamics through an integrated framework combining Earth system models, high-resolution river network topology, and lake bathymetry would advance understanding of how

climate–land–human interactions modulate water shortage propagation in atmosphere–basin–lake systems (Baydaroğlu et al., 2023; Shen, 2018). This study emphasizes the water transfer processes from the atmosphere to the basin and to the lake, so a tentative classification was conducted toward this objective. More classifications can be designed toward different goals.

## 5. Conclusions

This study investigated the propagation from AWD to lake drought in 1,617 lake basins of China and proposed a novel lake basin classification framework for lake drought management for the first time. This study emphasized the water shortage propagation from the atmosphere to the basin and to the lake, highlighting the importance of simultaneously addressing the adverse effects of climate change and strengthening lake management. The finding of this study extends current hydrological research of lakes, providing new insight into coping with lake drought under climate change.

## **Data Availability Statement**

We are grateful to Lange et al. (2023) for providing the ISIMIP3a precipitation dataset, Gosling et al. (2024) for providing the ISIMIP3a runoff dataset, Zhao (2021) for providing the lake water surface area dataset, Song et al. (2022a) for providing the geospatial database of nearly 100,000 reservoirs in China, Wang and Wang (2022b) for providing the GDP, and Wang and Wang (2022a) for providing the population dataset, and Xu et al. (2019) for providing the land use dataset.

## References

- AghaKouchak, A., Mirchi, A., Madani, K., Di Baldassarre, G., Nazemi, A., Alborzi, A., et al. (2021). Anthropogenic drought: Definition, challenges, and opportunities. Wiley Online Library.
- Apurv, T., & Cai, X. (2021). Regional drought risk in the contiguous United States. *Geophysical Research Letters*, 48(5), e2020GL092200. https://doi.org/10.1029/2020gl092200
- Avramoski, O. (2004). The role of public participation and citizen involvement in lake basin management. https://citeseerx.ist.psu.edu/document? repid=rep1&type=pdf&doi=3b9dd718341a2181cbecb02bff60e888d444bef5 (Retrieved from 26 June 2024).
- Awange, J. L., Sharifi, M. A., Ogonda, G., Wickert, J., Grafarend, E. W., & Omulo, M. A. (2008). The falling lake Victoria water level: GRACE, TRIMM and CHAMP satellite analysis of the lake basin. *Water Resources Management*, 22(7), 775–796. https://doi.org/10.1007/s11269-007-9191-y
- Baydaroğlu, Ö., Yeşilköy, S., Sermet, Y., & Demir, I. (2023). A comprehensive review of ontologies in the hydrology towards guiding next generation artificial intelligence applications. *Journal of Environmental Informatics*, 42(2), 90–107.
- Bedford, T., & Cooke, R. M. (2002). Vines—A new graphical model for dependent random variables. *Annals of Statistics*, 30(4), 1031–1068. https://doi.org/10.1214/aos/1031689016
- Cai, Z., Jin, T., Li, C., Ofterdinger, U., Zhang, S., Ding, A., & Li, J. (2016). Is China's fifth-largest inland lake to dry-up? Incorporated hydrological and satellite-based methods for forecasting Hulun lake water levels. Advances in Water Resources, 94, 185–199. https://doi.org/10.1016/j. advwatres.2016.05.010
- Cheng, J., & You, Z. (2019). Scientific connotation and practical paths about the principle of 'taking mountains, rivers, forests, farmlands, lakes, and grasslands as a life community' (in Chinese). China Population, Resources and Environment, 29(2), 1–6.
- Cooley, S. W., Ryan, J. C., & Smith, L. C. (2021). Human alteration of global surface water storage variability. *Nature*, 591(7848), 78–81. https://doi.org/10.1038/s41586-021-03262-3
- Cui, B.-L., & Li, X.-Y. (2016). The impact of climate changes on water level of Qinghai Lake in China over the past 50 years. *Hydrology Research*, 47(2), 532–542. https://doi.org/10.2166/nh.2015.237
- Fan, X. W., Zhang, Y., Shi, K., Peng, J., Liu, Y., Zhou, Y., et al. (2024). Surging compound drought-heatwaves underrated in global soils. Proceedings of the National Academy of Sciences of the United States of America, 121(42). https://doi.org/10.1073/pnas.2410294121
- Frieler, K., Volkholz, J., Lange, S., Schewe, J., Mengel, M., Del Rocío Rivas López, M., et al. (2024). Scenario setup and forcing data for impact model evaluation and impact attribution within the third round of the Inter-Sectoral Impact Model Intercomparison Project (ISIMIP3a). *Geoscientific Model Development*, 17(1), 1–51. https://doi.org/10.5194/gmd-17-1-2024
- Gaupp, F., Hall, J., Hochrainer-Stigler, S., & Dadson, S. (2020). Changing risks of simultaneous global breadbasket failure. Nature Climate Change, 10(1), 54–57. https://doi.org/10.1038/s41558-019-0600-z
- Gosling, S. N., et al. (2024). ISIMIP3a simulation data from the global water sector (v1.3) [dataset]. *ISIMIP Repository*. https://doi.org/10.48364/ ISIMIP.398165.398163
- Guttman, N. B. (1999). Accepting the standardized precipitation index: A calculation algorithm 1. JAWRA Journal of the American Water Resources Association, 35(2), 311–322. https://doi.org/10.1111/j.1752-1688.1999.tb03592.x
- Huang, Y., Yao, B., Li, Y., Zhang, H., & Wang, S. (2023). Deciphering Hulun lake level dynamics and periodical response to climate change during 1961–2020. Journal of Hydrology: Regional Studies, 46, 101352. https://doi.org/10.1016/j.ejrh.2023.101352
- Hughes, J., Petrone, K., & Silberstein, R. (2012). Drought, groundwater storage and stream flow decline in southwestern Australia. *Geophysical Research Letters*, 39(3). https://doi.org/10.1029/2011gl050797
- Jiang, T., Su, X., Zhang, G., Zhang, T., & Wu, H. (2023). Estimating propagation probability from meteorological to ecological droughts using a hybrid machine learning copula method. *Hydrology and Earth System Sciences*, 27(2), 559–576. https://doi.org/10.5194/hess-27-559-2023
- Kole, E., Koedijk, K., & Verbeek, M. (2007). Selecting copulas for risk management. Journal of Banking & Finance, 31(8), 2405–2423. https:// doi.org/10.1016/j.jbankfin.2006.09.010

#### Acknowledgments

This work was supported by the National Natural Science Foundation of China (42171037), the Youth Innovation Promotion Association of the Chinese Academy of Sciences (2021314), and the Jiangxi Science and Technology Program Project (20244BCF61001).

- Kong, L., Zheng, H., & Ouyang, Z. (2019). Ecological protection and restoration of forest, wetland, grassland and cropland based on the perspective of ecosystem services: A case study in Dongting lake watershed. Acta Ecologica Sinica, 39(23), 8903–8910. https://doi.org/10. 5846/stxb201905301137
- Kummu, M., Tes, S., Yin, S., Adamson, P., Józsa, J., Koponen, J., et al. (2014). Water balance analysis for the Tonle Sap Lake–floodplain system. *Hydrological Processes*, 28(4), 1722–1733. https://doi.org/10.1002/hyp.9718
- Lai, X., Jiang, J., Liang, Q., & Huang, Q. (2013). Large-scale hydrodynamic modeling of the middle Yangtze River Basin with complex river–lake interactions. Journal of Hydrology, 492, 228–243. https://doi.org/10.1016/j.jhydrol.2013.03.049
- Lai, X., Liang, Q., Jiang, J., & Huang, Q. (2014). Impoundment effects of the Three-Gorges-Dam on flow regimes in two China's largest freshwater lakes. Water Resources Management, 28(14), 5111–5124. https://doi.org/10.1007/s11269-014-0797-6
- Lange, S., Mengel, M., Treu, S., & Büchner, M. (2023). ISIMIP3a atmospheric climate input data (v1.2) [Dataset]. *ISIMIP Repository*. https://doi.org/10.48364/ISIMIP.982724.982722
- Li, X., Lin, Y., Ye, X., Yuan, C., Tan, Z., & Sun, T. (2024). The impacts of drought on the ecological niches of typical wetland plants in Poyang Lake, China. *Hydrology Research*, 55(9), nh2024033. https://doi.org/10.2166/nh.2024.033
- Liu, K., & Song, C. (2022). Modeling lake bathymetry and water storage from DEM data constrained by limited underwater surveys. Journal of Hydrology, 604, 127260. https://doi.org/10.1016/j.jhydrol.2021.127260
- Liu, K., Wang, J., Jiang, L., Richards, K. S., Sheng, Y., et al. (2021). Ongoing drainage reorganization driven by rapid lake growths on the Tibetan plateau. *Geophysical Research Letters*, 48(24). https://doi.org/10.1029/2021gl095795
- Long, D., Shen, Y. J., Sun, A., Hong, Y., Longuevergne, L., Yang, Y. T., et al. (2014). Drought and flood monitoring for a large karst plateau in Southwest China using extended GRACE data. *Remote Sensing of Environment*, 155, 145–160. https://doi.org/10.1016/j.rse.2014.08.006
- Long, D., Yang, W., Scanlon, B. R., Zhao, J., Liu, D., Burek, P., et al. (2020). South-to-North Water Diversion stabilizing Beijing's groundwater levels. *Nature Communications*, 11(1), 3665. https://doi.org/10.1038/s41467-020-17428-6
- Luo, S., Song, C., Ke, L., Zhan, P., Fan, C., Liu, K., et al. (2022). Satellite laser altimetry reveals a net water mass gain in global lakes with spatial heterogeneity in the early 21st century. *Geophysical Research Letters*, 49(3), 49. https://doi.org/10.1029/2021gl096676
- McKee, T. B., Doesken, N. J., & Kleist, J. (1993). The relationship of drought frequency and duration to time scales. Proceedings of the 8th Conference on Applied Climatology, 17, 179–183.
- Messager, M. L., Lehner, B., Grill, G., Nedeva, I., & Schmitt, O. (2016). Estimating the volume and age of water stored in global lakes using a geostatistical approach. *Nature Communications*, 7(1), 13603. https://doi.org/10.1038/ncomms13603
- Müller Schmied, H., Trautmann, T., Ackermann, S., Cáceres, D., Flörke, M., Gerdener, H., et al. (2023). The global water resources and use model WaterGAP v2. 2e: Description and evaluation of modifications and new features. *Geoscientific Model Development Discussions*, 2023, 1–46. Neath, A. A., & Cavanaugh, J. E. (2012). The Bayesian information criterion: Background, derivation, and applications. *Wiley Interdisciplinary*
- Reviews: Computational Statistics, 4(2), 199–203. https://doi.org/10.1002/wics.199
- Pekel, J.-F., Cottam, A., Gorelick, N., & Belward, A. S. (2016). High-resolution mapping of global surface water and its long-term changes. *Nature*, 540(7633), 418–422. https://doi.org/10.1038/nature20584
- Perales, K. M., Hein, C. L., Lottig, N. R., & Vander Zanden, M. J. (2020). Lake water level response to drought in a lake-rich region explained by lake and landscape characteristics. *Canadian Journal of Fisheries and Aquatic Sciences*, 77(11), 1836–1845. https://doi.org/10.1139/cjfas-2019-0270
- Peterson, T. J., Saft, M., Peel, M., & John, A. (2021). Watersheds may not recover from drought. *Science*, 372(6543), 745–749. https://doi.org/10. 1126/science.abd5085
- Qing, Y., Wang, S., Yang, Z.-L., & Gentine, P. (2023). Soil moisture–atmosphere feedbacks have triggered the shifts from drought to pluvial conditions since 1980. Communications Earth & Environment, 4(1), 254. https://doi.org/10.1038/s43247-023-00922-2
- Rachunok, B., & Fletcher, S. (2023). Socio-hydrological drought impacts on urban water affordability. *Nature Water*, 1(1), 83–94. https://doi.org/ 10.1038/s44221-022-00009-w
- Saber, A., James, D. E., & Hannoun, I. A. (2020). Effects of lake water level fluctuation due to drought and extreme winter precipitation on mixing and water quality of an alpine lake, Case Study: Lake Arrowhead, California. Science of the Total Environment, 714, 136762. https://doi.org/10. 1016/j.scitotenv.2020.136762
- Salvadore, E., Bronders, J., & Batelaan, O. (2015). Hydrological modelling of urbanized catchments: A review and future directions. *Journal of Hydrology*, 529, 62–81. https://doi.org/10.1016/j.jhydrol.2015.06.028
- Sattar, M. N., Lee, J. Y., Shin, J. Y., & Kim, T. W. (2019). Probabilistic characteristics of drought propagation from meteorological to hydrological drought in South Korea. Water Resources Management, 33(7), 2439–2452. https://doi.org/10.1007/s11269-019-02278-9
- Shen, C. (2018). A transdisciplinary review of deep learning research and its relevance for water resources scientists. *Water Resources Research*, 54(11), 8558–8593. https://doi.org/10.1029/2018wr022643
- Shukla, S., & Wood, A. W. (2008). Use of a standardized runoff index for characterizing hydrologic drought. Geophysical Research Letters, 35(2). https://doi.org/10.1029/2007g1032487
- Sklar, A. (1959). Fonctions de Repartition a n Dimensions et Leurs Marges. Publ.inst.statist.univ.paris.
- Soeprobowati, T. R. (2015). Integrated lake basin management for save Indonesian lake movement. *Procedia Environmental Sciences*, 23, 368–374. https://doi.org/10.1016/j.proenv.2015.01.053

Song, C., Fan, C., Zhu, J., Wang, J., Sheng, Y., Liu, K., et al. (2022a). A comprehensive geospatial database of nearly 100 000 reservoirs in China [Dataset]. Zenodo, 14(9), 4017–4034. https://doi.org/10.5281/zenodo.6984619

- Song, C., Fan, C., Zhu, J., Wang, J., Sheng, Y., Liu, K., et al. (2022b). A comprehensive geospatial database of nearly 100 000 reservoirs in China. *Earth System Science Data*, *14*(9), 4017–4034. https://doi.org/10.5194/essd-14-4017-2022
- Tong, Y., Feng, L., Wang, X., Pi, X., Xu, W., & Woolway, R. I. (2023). Global lakes are warming slower than surface air temperature due to accelerated evaporation. *Nature Water*, *1*(11), 1–12. https://doi.org/10.1038/s44221-023-00148-8
- Vacherat, A., Bonnet, S., & Mouthereau, F. (2018). Drainage reorganization and divide migration induced by the excavation of the Ebro basin (NE Spain). *Earth Surface Dynamics*, 6(2), 369–387. https://doi.org/10.5194/esurf-6-369-2018
- Vicente-Serrano, S. M., Pérez-Cabello, F., & Lasanta, T. (2008). Assessment of radiometric correction techniques in analyzing vegetation variability and change using time series of Landsat images. *Remote Sensing of Environment*, 112(10), 3916–3934. https://doi.org/10.1016/j.rse. 2008.06.011
- Wang, C., & Wang, J. (2022a). Kilometer grid dataset of China's historical population spatial distribution (1990-2015) [Dataset]. *National Tibetan Plateau*. https://doi.org/10.12078/2017121101
- Wang, C., & Wang, J. (2022b). Kilometer grid dataset of China's historical GDP spatial distribution (1990-2015) [Dataset]. National Tibetan Plateau. https://doi.org/10.12078/2017121102

- Wang, S., Zhang, L., She, D., Wang, G., & Zhang, Q. (2021). Future projections of flooding characteristics in the Lancang-Mekong River Basin under climate change. *Journal of Hydrology*, 602, 126778. https://doi.org/10.1016/j.jhydrol.2021.126778
- Wang, X. W., Shi, K., Qin, B. Q., Zhang, Y. L., & Woolway, R. I. (2024). Disproportionate impact of atmospheric heat events on lake surface water temperature increases. *Nature Climate Change*, 14(11), 1172–1177. https://doi.org/10.1038/s41558-024-02122-y
- Weyhenmeyer, G. A., Chukwuka, A. V., Anneville, O., Brookes, J., Carvalho, C. R., Cotner, J. B., et al. (2024). Global lake health in the Anthropocene: Societal implications and treatment strategies. *Earth's Future*, 12(4), e2023EF004387. https://doi.org/10.1029/2023ef004387
- Woolway, R. I., Kraemer, B. M., Lenters, J. D., Merchant, C. J., O'Reilly, C. M., & Sharma, S. (2020). Global lake responses to climate change. *Nature Reviews Earth & Environment*, 1(8), 388–403. https://doi.org/10.1038/s43017-020-0067-5
- Woolway, R. I., Verburg, P., Lenters, J. D., Merchant, C. J., Hamilton, D. P., Brookes, J., et al. (2018). Geographic and temporal variations in turbulent heat loss from lakes: A global analysis across 45 lakes. *Limnology & Oceanography*, 63(6), 2436–2449. https://doi.org/10.1002/lno. 10950
- Xi, Y., Peng, S., Ciais, P., & Chen, Y. (2021). Future impacts of climate change on inland Ramsar wetlands. Nature Climate Change, 11(1), 45–51. https://doi.org/10.1038/s41558-020-00942-2
- Xie, J., Liu, X., Bai, P., & Liu, C. (2022). Rapid watershed delineation using an automatic outlet relocation algorithm. *Water Resources Research*, 58(3), e2021WR031129. https://doi.org/10.1029/2021wr031129
- Xu, F., Zhang, G., Woolway, R. I., Yang, K., Wada, Y., Wang, J., & Crétaux, J.-F. (2024). Widespread societal and ecological impacts from projected Tibetan Plateau lake expansion. *Nature Geoscience*, 17(6), 1–8. https://doi.org/10.1038/s41561-024-01446-w
- Xu, K., Yang, D. W., Xu, X. Y., & Lei, H. M. (2015). Copula based drought frequency analysis considering the spatio-temporal variability in Southwest China. Journal of Hydrology, 527, 630–640. https://doi.org/10.1016/j.jhydrol.2015.05.030
- Xu, X., Liu, J., Zhang, S., Li, R., Yan, C., & Wu, S. (2019). Landuse dataset in China (1980-2015) [dataset]. National Tibetan Plateau. https://data.tpdc.ac.cn/zh-hans/data/a75843b75844-76591-75844a75869-a75845e75844-75846f94099ddc75842d/
- Yang, X. L., Wu, F., Yuan, S. S., Ren, L. L., Sheffield, J., Fang, X. Q., et al. (2024). Quantifying the impact of human activities on hydrological drought and drought propagation in China using the PCR-GLOBWB v2.0 model. Water Resources Research, 60(1). https://doi.org/10.1029/ 2023wr035443
- Yang, Y., McVicar, T. R., Donohue, R. J., Zhang, Y., Roderick, M. L., Chiew, F. H., et al. (2017). Lags in hydrologic recovery following an extreme drought: Assessing the roles of climate and catchment characteristics. *Water Resources Research*, 53(6), 4821–4837. https://doi.org/ 10.1002/2017wr020683
- Yang, Z. S., Yue, P., Zhang, Q., He, H., & Yang, H. W. (2025). Multi-land-surface variables-precipitation coupling over the northeastern slope of the Tibetan Plateau. *Theoretical and Applied Climatology*, 156(1), 62. https://doi.org/10.1007/s00704-024-05252-7
- Yao, F., Livneh, B., Rajagopalan, B., Wang, J., Crétaux, J.-F., Wada, Y., & Berge-Nguyen, M. (2023). Satellites reveal widespread decline in global lake water storage. *Science*, 380(6646), 743–749. https://doi.org/10.1126/science.abo2812
- Yao, J., Zhang, Q., Ye, X., Zhang, D., & Bai, P. (2018). Quantifying the impact of bathymetric changes on the hydrological regimes in a large floodplain lake: Poyang Lake. *Journal of Hydrology*, 561, 711–723. https://doi.org/10.1016/j.jhydrol.2018.04.035
- Zhang, D., Chen, P., Zhang, Q., & Li, X. (2017). Copula-based probability of concurrent hydrological drought in the Poyang lake-catchment-river system (China) from 1960 to 2013. Journal of Hydrology, 553, 773–784. https://doi.org/10.1016/j.jhydrol.2017.08.046
- Zhang, D., Zhang, Q., Qiu, J., Bai, P., Liang, K., & Li, X. (2018). Intensification of hydrological drought due to human activity in the middle reaches of the Yangtze River, China. Science of the Total Environment, 637, 1432–1442. https://doi.org/10.1016/j.scitotenv.2018.05.121
- Zhang, Q., Xue, C., & Xia, J. (2023). Impacts, contributing factors and countermeasures of extreme droughts in Poyang Lake. *Bulletin of Chinese Academy of Sciences (in Chinese)*, 38(12), 1894–1902.
- Zhang, Y., Li, C., Chiew, F. H., Post, D. A., Zhang, X., Ma, N., et al. (2023). Southern hemisphere dominates recent decline in global water availability. *Science*, 382(6670), 579–584. https://doi.org/10.1126/science.adh0716
- Zhao, G. (2021). Global lake evaporation volume (GLEV) dataset [Dataset]. Zenodo. https://doi.org/10.5281/zenodo.4646621

Zhao, G., & Gao, H. L. (2018). Automatic correction of contaminated images for assessment of reservoir surface area dynamics. *Geophysical Research Letters*, 45(12), 6092–6099. https://doi.org/10.1029/2018gl078343

- Zhao, G., Li, Y., Zhou, L., & Gao, H. (2022). Evaporative water loss of 1.42 million global lakes. *Nature Communications*, 13(1), 1–10. https://doi.org/10.1038/s41467-022-31125-6
- Zheng, J., Sun, G., Li, W., Yu, X., Zhang, C., Gong, Y., & Tu, L. (2016). Impacts of land use change and climate variations on annual inflow into the Miyun Reservoir, Beijing, China. Hydrology and Earth System Sciences, 20(4), 1561–1572. https://doi.org/10.5194/hess-20-1561-2016
- Zhu, H., Chen, K. J., Hu, S. Q., Liu, J. G., Shi, H. Y., Wei, G. G., et al. (2023). Using the global navigation satellite system and precipitation data to Establish the propagation characteristics of meteorological and hydrological drought in Yunnan, China. Water Resources Research, 59(4), e2022WR033126. https://doi.org/10.1029/2022wr033126